Kidney Diseases

# **Research Article**

Kidney Dis 2024;10:249–261 DOI: 10.1159/000538929 Received: December 5, 2023 Accepted: April 11, 2024 Published online: April 16, 2024

# Impact of Arterial Calcification on Cardiovascular and Renal Outcomes in Kidney Transplant Patients

Joohyung Ha<sup>a</sup> Jong Cheol Jeong<sup>b</sup> Jung-Hwa Ryu<sup>c</sup> Myung-Gyu Kim<sup>d</sup> Kyu Ha Huh<sup>e</sup> Kyo Won Lee<sup>f</sup> Hee-Yeon Jung<sup>g</sup> Kyung Pyo Kang<sup>h</sup> Han Ro<sup>i</sup> Seungyeup Han<sup>j</sup> Beom Seok Kim<sup>a</sup> Jaeseok Yang<sup>a</sup> on behalf of the KNOW-KT Study Group

<sup>a</sup>Department of Internal Medicine, Yonsei University College of Medicine, Severance Hospital, Seoul, Republic of Korea; <sup>b</sup>Department of Internal Medicine, Seoul National University College of Medicine, Bundang Hospital, Seongnam, Republic of Korea; <sup>c</sup>Department of Internal Medicine, Ewha Womans University Medical Center, Seoul, Republic of Korea; <sup>d</sup>Department of Internal Medicine, Korea University College of Medicine, Seoul, Republic of Korea; <sup>e</sup>Department of Transplantation Surgery, Severance Hospital, Yonsei University Health System, Seoul, Republic of Korea; <sup>f</sup>Department of Surgery, Sungkyunkwan University, Seoul Samsung Medical Center, Seoul, Republic of Korea; <sup>g</sup>Department of Internal Medicine, Kyungpook National University Hospital, Daegu, Republic of Korea; <sup>h</sup>Department of Internal Medicine, Jeonbuk National University Medical School, Jeonju, Republic of Korea; <sup>i</sup>Department of Internal Medicine, Gachon University, Gil Hospital, Incheon, Republic of Korea; <sup>j</sup>Department of Internal Medicine, Gachon Medical Center, Daegu, Republic of Korea;

#### **Keywords**

Aortic artery calcification · Cardiovascular disease · Coronary artery calcification · Kidney transplantation · Renal outcome

## Abstract

**Introduction:** Coronary artery calcification score (CACS) and abdominal aortic calcification score (AACS) are both wellestablished markers of vascular stiffness, and previous studies have shown that a higher CACS is a risk factor for chronic kidney disease (CKD) progression. However, the impact of pretransplant CACS and AACS on cardiovascular and renal outcomes in kidney transplant patients has not been established. **Methods:** We included 944 kidney transplant recipients from the KoreaN cohort study for

karger@karger.com www.karger.com/kdd

Karger<sup>\*</sup>

**∂OPEN ACCESS** 

© 2024 The Author(s). Published by S. Karger AG, Basel

This article is licensed under the Creative Commons Attribution-NonCommercial 4.0 International License (CC BY-NC) (http://www. karger.com/Services/OpenAccessLicense). Usage and distribution for commercial purposes requires written permission. Outcome in patients With Kidney Transplantation (KNOW-KT) cohort and categorized them into three groups (low, medium, and high) according to baseline CACS (0, 0 < and ≤100, >100) and AACS (0, 1–4, >4). The low (0), medium  $(0 < and \le 100)$ , and high (>100) CACS groups each consisted of 462, 213, and 225 patients, respectively. Similarly, the low (0), medium (1-4), and high (>4) AACS groups included 638, 159, and 147 patients, respectively. The primary outcome was the occurrence of cardiovascular events. The secondary outcomes were all-cause mortality and composite kidney outcomes, which comprised of >50% decline in the estimated glomerular filtration rate and graft loss. Cox regression analysis was used to investigate the association between baseline CACS/AACS and outcomes. Results: The high CACS group (N = 462) faced a significantly higher risk for cardiovascular outcomes (adjusted hazard ratio [aHR], 5.97;

Correspondence to: Jaeseok Yang, jcyjs@yuhs.ac 95% confidence interval [CI], 2.01–17.7) and all-cause mortality (aHR, 2.74; 95% CI, 1.27–5.92) compared to the low CACS group (N = 225). Similarly, the high AACS group (N = 638) had an elevated risk for cardiovascular outcomes (aHR, 2.38; 95% CI, 1.16–4.88). Furthermore, the addition of CACS to prediction models improved prediction indices for cardiovascular outcomes. However, the risk of renal outcomes did not differ among CACS or AACS groups. **Conclusion:** Pretransplant arterial calcification, characterized by high CACS or AACS, is an independent risk factor for cardiovascular outcomes and mortality in kidney transplant patients. © 2024 The Author(s).

Published by S. Karger AG, Basel

### **Plain Language Summary**

Arterial calcification, accumulation of calcium in the arterial walls, vascular stiffness, and loss of elasticity of blood vessels can contribute to the development of cardiovascular diseases. Patients with chronic kidney disease and those undergoing dialysis have a considerably increased risk of vascular calcification. Even after kidney transplantation when kidney function has been restored, the prevalence of vascular calcification and subsequent cardiovascular disease remains high. Coronary artery calcification score and abdominal aortic calcification score are both well-established markers of vascular calcification. However, the impact of pretransplant vascular calcification scores on cardiovascular and renal outcomes in kidney transplant patients has not been established. When we analyzed 944 Korean kidney transplant patients, both vascular calcification scores were significantly associated with cardiovascular outcomes after kidney transplantation, but were not associated with renal outcomes. We also demonstrated that the addition of coronary artery calcification scores led to a modest improvement in the prediction performance for kidney transplant outcomes. Our findings suggest a potential role of pretransplant screening of coronary calcification scores and aortic calcification scores in risk stratification for post-kidney transplant outcomes. © 2024 The Author(s).

© 2024 The Author(s). Published by S. Karger AG, Basel

### Introduction

Cardiovascular disease (CVD) is a leading cause of morbidity and mortality among kidney transplant (KT) patients [1–3]. Arterial calcification, accumulation of calcium in the arterial walls, vascular stiffness, and loss of elasticity of blood vessels can contribute to the development of CVD [4].

Patients with chronic kidney disease (CKD) and those undergoing dialysis have a considerably increased risk of vascular calcification due to various factors, including inflammation, abnormal bone mineral metabolism, and oxidative stress [5, 6]. Chronic imbalances in serum calcium, phosphate, and parathyroid hormone levels in patients with CKD may also play a role by disrupting the equilibrium between bone formation and resorption, leading to ectopic mineralization in the systemic vasculature [7–9]. Notably, coronary artery calcification score (CACS) and abdominal aortic calcification score (AACS) are two well-established markers that can objectively quantify vascular calcification and arterial stiffness [10–14]. Higher CACS is a risk factor for CVD and CKD progression in patients with CKD [15, 16]. Additionally, higher AACS is also a risk factor for CVD and CKD progression in patients with CKD [17, 18].

KT patients are, by definition, a subgroup of CKD patients with a single functional kidney. Even after kidney function has been restored, the presence of traditional and nontraditional risk factors and transplant-related factors, such as immunosuppressive medications, leads to a high prevalence of vascular calcification and subsequent CVD in KT recipients [19]. Therefore, understanding the pathogenic mechanisms that leads to vascular lesions in KT recipients may be crucial for optimizing long-term risk management after KT. However, only a few single-center studies with a small sample size have examined the impact of higher CACS and AACS on CVD, but not on CKD progression in the KT population [20-24]. We aimed to investigate the impact of arterial calcification and vascular stiffness on cardiovascular and renal outcomes in KT patients.

### Methods

#### Study Design and Participants

The KoreaN cohort study for Outcome in patients With Kidney Transplantation (KNOW-KT) is a prospective, multicenter, observational cohort study in Korean KT patients [25]. The study enrolled 1,080 KT patients from 8 transplant centers between 2012 and 2016 who voluntarily provided written informed consent. We excluded 56 patients with no available baseline CACS or AACS. Among the 1,080 patients, 125 were excluded due to withdrawal of consent or a participating center. Additionally, 55 and 11 were excluded due to a lack of CACS and AACS, respectively. The final analyses for CACS and AACS included 900 and 944 patients, respectively (online suppl. Fig. S1; for all online suppl. material, see https://doi.org/10.1159/000538929). This study was conducted in accordance with the principles of the Declaration of Helsinki and the Declaration of Istanbul. The Regional Ethics Committee and Institutional Review Board of each participating center approved the study protocol.

### Data Collection and Measurements

The following baseline clinical characteristics were collected at the time of KT: (1) donor information (age, sex, body weight, height, comorbidities, history of malignancy); (2) recipient information (age, sex, smoking history, cause of end-stage kidney disease, history of malignancy and CVDs, the age-adjusted Charlson Comorbidity Index (CCI), medications); and (3) KT-related information (date of transplantation, prior transplant history, donor-recipient relationship, and desensitization). The CCI is a widely adopted method to assess comorbid conditions, such as myocardial infarction, congestive heart failure, and diabetes mellitus (DM), and its total score can be used to estimate 10-year mortality [26].

The pretransplant cardiovascular evaluations included echocardiography, measurement of pulse wave velocity and ankle brachial index, CACS based on computed tomography, and AACS based on simple radiographs of the lateral lumbar spine using the Kauppila score, a validated scoring system ranging from 0 to 24 points [13]. CACS was measured at baseline and 5-year follow-up, while AACS was measured at baseline, 3-year, and 5-year follow-up. Patients were divided into three groups (low [0], medium [0 < and ≤100], and high [>100]) according to baseline CACS values and were divided into three groups (low [0], medium [1–4], and high [>4]) according to baseline AACS values, which was in accordance with categorization of CACS and AACS in previous studies [16, 27]. Laboratory data and clinical events were collected and documented annually.

#### Study Outcomes

The primary outcome was the occurrence of cardiovascular events. The secondary outcomes were all-cause mortality and composite kidney outcomes, which comprised of renal function decline and graft loss. Myocardial infarction, coronary revascularization, stroke, and new-onset or aggravation of congestive heart failure were considered cardiovascular events. Renal function decline was defined as a >50% decline in the estimated glomerular filtration rate (eGFR) compared to the baseline value measured <28 days after KT, whereas graft loss was defined as death-censored graft loss requiring maintenance dialysis for more than 3 months or re-transplantation. Serum creatinine levels were measured using an isotope-dilution mass spectrometry-traceable method, and eGFR was calculated using the Chronic Kidney Disease Epidemiology Collaboration creatinine equation [28].

#### Statistical Analysis

Demographic and laboratory parameters were summarized using percentages for categorical variables and using mean, median, and standard deviations for continuous variables, as appropriate. The Wilcoxon signed-rank test was used to analyze difference in CACS or AACS between time intervals. Spearman's rank correlation test was used to analyze the correlation of CACS or AACS between the baseline and post-KT 5-year. We used Cox regression analysis to investigate the association between baseline CACS/AACS and post-KT outcomes, including cardiovascular events, composite kidney outcomes, and all-cause mortality. Model 1 was unadjusted, univariate analysis, and model 2 was adjusted for the CCI, smoking status, education level, body mass index, and cause of end-stage kidney disease. Model 3 included KT-specific covariates such as the type of donor (living or deceased), duration of dialysis before KT, desensitization, and immunosuppression medications. All relevant covariates had <10% missingness, and multiple imputations by chained equations were used to fill in the missing values [29]. The results of the multivariate regression analysis are presented as hazard ratios (HRs) and 95% confidence intervals (CIs).

Harrell's C-statistics and net reclassification improvement (NRI) were calculated for the prediction models to assess the additive value of the CACS or AACS in the Cox regression models [30]. The bootstrapping method was used to derive 95% CIs. Model 3 was used as the comparator for the base model.

Additionally, risk factors for the progression of vascular calcification during 5 years were analyzed. We defined CACS progression as a binary outcome, which was positive when the square root of the difference between baseline and 5-year follow-up CACS was greater than 2.5, as a previous study reported [31]. Similarly, AACS progression was defined as positive when either 3-year follow-up AACS or 5-year follow-up AACS was numerically greater than the baseline AACS. We then conducted multivariable analyses using the generalized estimating equation and calculated the odds ratios with 95% CIs. Statistical analysis was performed using the statistical software R (R Foundation for Statistical Computing, Vienna, Austria). p < 0.05 was considered to be statistically significant.

### Results

# Baseline Clinical Characteristics

The baseline clinical characteristics of the study population are presented according to three CACS and three AACS groups in Tables 1 and 2, respectively. Laboratory findings of the study population are also presented according to three CACS and three AACS groups in online supplementary Table S1 and S2, respectively. In the CACS analysis, the median age was 47 years (interquartile range: 18-71 years), and 62.3% were men. The median CACS value was 0.0 (minimum 0.0, maximum 8,888.0). In the AACS analysis, the median age was 48 years (interguartile range: 17-71 years), and 62.3% were men. The median AACS value was 0.0 (minimum 0.0, maximum 21.0). Patients with higher arterial calcification scores (CACS or AACS) were more likely to be male (p < 0.001), older (p < 0.001) 0.001), and more likely to have DM at baseline (p < 0.001). Hypertension was more common in patients with a higher CACS (p = 0.002). Cardiovascular comorbidities, such as coronary artery disease and cerebrovascular disease, were more common in the groups with higher CACS or AACS than in those with lower CACS or AACS (Tables 1, 2).

# Posttransplant Changes in CACS and AACS

CACS and AACS increased slightly in most patients, even after KT (Fig. 1a, b). The mean CACS (p < 0.001) and AACS (p < 0.001) at 5 years after KT were higher than the pretransplant baseline levels. However, in the

	Total ( $N = 900$ )	0 ( <i>N</i> = 462)	$0 < and \le 100 (N = 213)$	>100 (N = 225)	p value
CACS	0.0 [0.0, 8888.0]	0.0 [0.0, 0.0]	19.3 [0.1, 99.2]	461.7 [104.0, 8888.0]	<0.001
Type of KT, <i>n</i> (%) LDKT DDKT	729 (81.0) 171 (19.0)	415 (89.8) 47 (10.2)	164 (77.0) 49 (23.0)	150 (66.7) 75 (33.3)	<0.001
Sex (male), n (%) Age, years BMI, kg/m <sup>2</sup> Waist-hip ratio Systolic BP, mm Hg Diastolic BP, mm Hg CCI	561 (62.3) 47.0 [18.0, 71.0] 22.6 [14.7, 35.7] 0.89 [0.70, 1.17] 137 [90.0, 215] 83.0 [51.0, 123] 2.0 [0.0, 8.0]	248 (53.7) 41.0 [18.0, 69.0] 21.6 [14.7, 34.8] 0.88 [0.70, 1.17] 134 [90.0, 210] 83.0 [54.0, 123] 2.0 [0.0, 6.0]	142 (66.7) 50.0 [27.0, 70.0] 23.2 [15.9, 34.6] 0.91 [0.74, 1.07] 142 [100, 215] 84.0 [57.0, 115] 2.0 [0.0, 6.0]	171 (76.0) 54.0 [31.0, 71.0] 23.8 [17.2, 35.7] 0.92 [0.75, 1.13] 139 [99.0, 191] 80.0 [51.0, 113] 3.0 [0.0, 8.0]	<0.001 <0.001 <0.001 <0.001 <0.001 0.007 <0.001
Etiology of ESKD, n (%) DM Hypertension Glomerulonephritis Others	184 (20.4) 197 (21.9) 277 (30.8) 242 (26.9)	45 (9.7) 95 (20.6) 190 (41.1) 132 (28.6)	44 (20.7) 52 (24.4) 54 (25.4) 63 (29.5)	95 (42.2) 50 (22.2) 33 (14.7) 47 (20.9)	<0.001
Smoking, <i>n</i> (%) Never Current Former	475 (52.8) 67 (7.4) 358 (39.8)	280 (60.6) 25 (5.4) 157 (34.0)	100 (46.9) 16 (7.5) 97 (45.5)	95 (42.2) 26 (11.6) 104 (46.2)	<0.001
Dialysis before KT, n (%) HD PD KT Preemptive	589 (65.4) 110 (12.2) 15 (1.7) 186 (20.7)	287 (62.1) 40 (8.7) 7 (1.5) 128 (27.7)	130 (61.0) 39 (18.3) 1 (0.5) 43 (20.2)	172 (76.4) 31 (13.8) 7 (3.1) 15 (6.7)	<0.001
DM Hypertension Congestive heart failure Arrhythmia Coronary artery disease Cerebrovascular disease	225 (25.0) 832 (92.4) 20 (2.2) 13 (1.4) 50 (5.6) 37 (4.1)	54 (11.7) 405 (87.7) 9 (1.9) 4 (0.9) 8 (1.7) 9 (1.9)	57 (26.8) 208 (97.7) 5 (2.3) 6 (2.8) 7 (3.3) 9 (4.2)	114 (50.7) 219 (97.3) 6 (2.7) 3 (1.3) 35 (15.6) 19 (8.4)	<0.001 0.002 0.889 0.163 <0.001 <0.001

Table 1. Baseline characteristics according to CACS

Note: All continuous variables are expressed as median [min, max]. All categorical variables are expressed as number and percentage. CACS, coronary artery calcium score; KT, kidney transplantation; LDKT, living donor kidney transplantation; DDKT, deceased donor kidney transplantation; BMI, body mass index; BP, blood pressure; ESKD, end-stage kidney disease; HD, he-modialysis; PD, peritoneal dialysis.

low CACS (CACS 0) and AACS (AACS 0) subgroups, significant proportions (52% in CACS, 42% in AACS) still had undetectable vascular calcification (CACS 0, AACS 0) at the 5-year after KT. When the progression of CACS/ AACS was analyzed according to the subgroups of CACS/ AACS, the progression was statistically significant in all of the subgroups, except in the high AACS (AACS >4) subgroup (online suppl. Fig. S2, S3). A significant correlation was observed between pretransplant CACS and 5-year CACS ( $r^2 = 0.835$ ; p < 0.001; Fig. 1c). The correlation between pretransplant and 5-year AACSs was also significant ( $r^2 = 0.526$ ; p < 0.001; Fig. 1d).

When risk factors for the progression of vascular calcification were assessed, old age, DM, and high body mass index were significant risk factors for CACS progression. For AACS, old age, DM, and high serum calcium levels were independently associated with the risk of AACS progression (online suppl. Table S3).

# Cardiovascular Outcomes and All-Cause Mortality

At a median follow-up of 7.8 years, the incidence of cardiovascular outcomes was 1.5, 11.1, and 20.4 per 1,000 person-years in the low, medium, and high CACS groups, respectively (Table 3). The all-cause mortality

	Overall ( $N = 944$ )	0 ( <i>N</i> = 638)	1–4 ( <i>N</i> = 159)	>4 ( <i>N</i> = 147)	p value
AACS	0.0 [0.0, 21.0]	0.0 [0.0, 0.0]	2.0 [1.0, 4.0]	9.0 [5.0, 21.0]	<0.001
Type of KT, <i>n</i> (%) LDKT DDKT	759 (80.4) 185 (19.6)	540 (84.6) 98 (15.4)	121 (76.1) 38 (23.9)	98 (66.7) 49 (33.3)	<0.001
Sex (male), n (%) Age, years BMI, kg/m <sup>2</sup> Waist-hip ratio Systolic BP (mm Hg) Diastolic BP, mm Hg CCI	588 (62.3) 48.0 [17.0, 71.0] 22.6 [14.7, 39.0] 0.90 [0.70, 1.21] 136 [90.0, 215] 82.0 [51.0, 123] 2.0 [0.0, 8.0]	372 (58.3) 44.0 [17.0, 70.0] 22.3 [14.7, 39.0] 0.89 [0.71, 1.17] 135 [90.0, 215] 84.0 [54.0, 120] 2.0 [0.0, 6.0]	117 (73.6) 52.0 [26.0, 71.0] 22.8 [17.1, 33.7] 0.90 [0.74, 1.10] 138 [95.0, 191] 80.0 [51.0, 123] 2.0 [0.0, 8.0]	99 (67.3) 54.0 [26.0, 71.0] 23.8 [15.5, 34.8] 0.91 [0.70, 1.21] 140 [100, 187] 80.0 [51.0, 112] 3.0 [1.0, 7.0]	<0.001 <0.001 <0.001 <0.001 0.009 <0.001 <0.001
Etiology of ESKD, n (%) DM Hypertension Glomerulonephritis Others	192 (20.3) 215 (22.8) 295 (31.3) 242 (25.6)	77 (12.1) 152 (23.8) 240 (37.6) 169 (26.5)	52 (32.7) 36 (22.6) 34 (21.4) 37 (23.3)	63 (42.9) 27 (18.4) 21 (14.3) 36 (24.4)	<0.001
Smoking, <i>n</i> (%) Never Current Former	519 (55.0) 67 (7.1) 358 (37.9)	373 (58.5) 31 (4.9) 234 (36.7)	68 (42.8) 21 (13.2) 70 (44.0)	78 (53.1) 15 (10.2) 54 (36.7)	<0.001
Dialysis before KT, n (%) HD PD KT Preemptive	611 (64.7) 119 (12.6) 14 (1.5) 200 (21.2)	395 (61.9) 78 (12.2) 7 (1.1) 158 (24.8)	111 (69.8) 18 (11.3) 2 (1.3) 28 (17.6)	105 (71.4) 23 (15.6) 5 (3.4) 14 (9.5)	<0.001
DM Hypertension Congestive heart failure Arrhythmia Coronary artery disease Cerebrovascular disease	242 (25.6) 867 (91.8) 19 (2.0) 14 (1.5) 55 (5.8) 36 (3.8)	100 (15.7) 574 (90.0) 9 (1.4) 10 (1.6) 16 (2.5) 16 (2.5)	61 (38.4) 153 (96.2) 5 (3.1) 2 (1.3) 15 (9.4) 10 (6.3)	81 (55.1) 140 (95.2) 5 (3.4) 2 (1.4) 24 (16.3) 10 (6.8)	<0.001 0.483 0.196 0.925 <0.001 0.020

Table 2. Baseline characteristics according to AACS

Note: All continuous variables are expressed as median [min, max]. All categorical variables are expressed as number and percentage. AACS, abdominal aortic calcium score; KT, kidney transplantation; LDKT, living donor kidney transplantation; DDKT, deceased donor kidney transplantation; BMI, body mass index; BP, blood pressure; ESKD, end-stage kidney disease; HD, he-modialysis; PD, peritoneal dialysis.

rates were 1.5, 4.6, and 8.9 per 1,000 person-years in the low, medium, and high CACS groups, respectively. The Kaplan-Meier survival analysis showed cumulative incidence curves for each CACS group (Fig. 2). Significant differences were observed in cardiovascular outcomes (p < 0.001, Fig. 2a) and allcause mortality (p < 0.001, Fig. 2b) among the different CACS groups according to the log-rank test. In the Cox regression analysis, the risks for cardiovascular events and all-cause mortality in KT recipients increased as CACS increased. Higher CACS as a continuous variable was associated with a higher risk for cardiovascular events (HR: 1.004, 95% CI: 1.004–1.005, p < 0.001, model 3) and all-cause mortality (HR: 1.003, 95% CI: 1.002–1.003, p < 0.001, model 3, Table 3). The high CACS group (CACS >100) had a higher risk for cardiovascular events (HR: 5.97, 95% CI: 2.01–17.7, p = 0.002, model 3) and all-cause mortality (HR: 2.74: 95% CI: 1.27–5.92, p = 0.009, model 3) than the low CACS group (Table 3). The medium CACS group (0 <CACS ≤100) also had a higher risk for cardiovascular events (HR: 4.35, 95% CI: 1.52–12.5, p = 0.007, model 3) than the low CACS group (Table 3).



Fig. 1. Changes in CACS and AACS after kidney transplantation. a Changes in CACS over 5 years after kidney transplantation. \*\* indicates *p* < 0.001 compared to the baseline values (Wilcoxon signedrank test). b Changes in AACS over 5 years after kidney transplantation. Each value indicates mean with the standard error of the mean. \*\* indicates p < 0.001 compared to the baseline values. ## indicates p = 0.007between the 3-year and 5-year AACS values (Wilcoxon signed-rank test). c Correlation between CACS values at baseline and 5-year follow-up after kidney transplantation. p < 0.001 (Spearman's rank correlation test). d Correlation between AACS values at baseline and 5-year follow-up after kidney transplantation. CACS, coronary artery calcification score; AACS, aortic artery calcification score. p <0.001 (Spearman's rank correlation test).

Outcomes	CACS	Events	Person- years	Incidence (/1,000 person- years)	HR Model 1ª	HR Model 2 <sup>b</sup>	HR Model 3 <sup>c</sup>	p value Model 3 <sup>c</sup>
Cardiovascular outcome	0 0 < and ≤100 >100	5 16 29	3,395 1,440 1,420	1.5 11.1 20.4	Reference 7.37 (2.70–20.1) 13.25 (5.13–34.2)	Reference 4.21 (1.48–12.0) 5.62 (2.01–15.7)	Reference 4.35 (1.52–12.5) 5.97 (2.01–17.7)	_ 0.007 0.002
		HR increase per 1 CACS			1.005 (1.004–1.006)	1.004 (1.004–1.005)	1.004 (1.004–1.005)	<0.001
All-cause mortality	0 0 < and ≤100 >100	5 7 14	3,427 1,524 1,570	1.5 4.6 8.9	Reference 3.18 (1.01–10.0) 6.18 (2.23–17.2)	Reference 1.88 (0.55–6.45) 2.99 (0.90–9.88)	Reference 1.74 (0.73–4.14) 2.74 (1.27–5.92)	– 0.207 0.009
		HR increase per 1 CACS			1.004 (1.003–1.005)	1.003 (1.003–1.004)	1.003 (1.002–1.003)	<0.001
Renal outcome	0 0 < and ≤100 >100	70 39 39	3,145 1,368 1,424	22.3 28.5 27.4	Reference 1.27 (0.86–1.88) 1.22 (0.83–1.81)	Reference 1.17 (0.77–1.80) 1.04 (0.65–1.66)	Reference 1.11 (0.72–1.72) 0.94 (0.57–1.54)	– 0.649 0.807
		HR inci	rease per	1 CACS	1.001 (0.988–1.013)	1.000 (0.979–1.021)	1.000 (0.994–1.004)	0.166

Table 3. Incidence and HRs for the outcomes according to CACS

The HR (95% CI) is reported for each model. The cardiovascular outcome was defined as myocardial infarction, coronary revascularization, stroke and new-onset or aggravation of congestive heart failure. The renal outcome was defined as the first occurrence of a more than 50% decline in eGFR compared to a baseline value measured less than 28 days after kidney transplantation, or death-censored graft loss requiring maintenance dialysis for more than 3 months, or re-transplantation. CACS, coronary artery calcification score; HR, hazard ratio. <sup>a</sup>Model 1: a crude analysis without adjustment. <sup>b</sup>Model 2: adjusted for CCI, smoking status, education levels, body mass index, and cause of end-stage kidney disease. <sup>c</sup>Model 3: Model 2 + type of donor (living or deceased), duration of dialysis before kidney transplantation, desensitization, and immunosuppression medications.

In the AACS analysis, the incidence of cardiovascular outcomes was 4.5, 16.6, and 23.5 per 1,000 person-years in the low, medium, and high AACS groups, respectively (Table 4). The Kaplan-Meier survival analysis showed cumulative incidence curves for each AACS group (Fig. 3). Significant differences were observed in cardiovascular outcomes (p < 0.001, Fig. 3a) and all-cause mortality (p =0.001, Fig. 3b) among different AACS groups according to the log-rank test. Higher AACS as a continuous variable was associated with a higher risk for cardiovascular events (HR: 1.08, 95% CI: 1.02–1.14, *p* = 0.002, model 3) and showed a tendency of association with all-cause mortality (HR: 1.05, 95% CI: 0.95–1.18, *p* = 0.109, model 3, Table 4). Both the high AACS group (AACS >4, HR: 2.38, 95% CI: 1.16-4.88, p = 0.028, model 3) and the medium AACS group (AACS = 1-4, HR: 2.12, 95% CI: 1.05-4.27, p = 0.049,

model 3) had a higher risk for cardiovascular events (Table 4). Moreover, the high AACS group (AACS >4, HR: 1.89, 95% CI: 0.70–5.06, p = 0.179, model 3) and the medium AACS group (AACS = 1–4, HR: 2.13, 95% CI: 0.82–5.51, p = 0.163, model 3) showed a tendency of increased risk for all-cause mortality compared to the low AACS group, despite no statistical significance (Table 4).

# Renal Outcomes

The risk of renal outcomes, which we defined as a >50% decline in the eGFR compared to the baseline value measured less than 28 days after the KT operation or graft loss, did not differ significantly among the different CACS groups (p = 0.403, Fig. 2c). The HR for the high CACS group compared to the low CACS group was 0.94 (95% CI: 0.57–1.54, p = 0.807, model 3, Table 3).



**Fig. 2.** Kaplan-Meier curves for incidence of cardiovascular outcomes (**a**), all-cause mortality (**b**), and renal outcomes (**c**) according to CACS groups (CACS = 0; 0 < CACS  $\leq$  100, CACS > 100). *p* < 0.001, <0.001, and 0.403 for cardiovascular outcomes, all-cause mortality, and renal outcomes, respectively (log-rank test). CACS, coronary artery calcification score.

Outcomes	AACS	Events	Person- years	Incidence (/1,000 person-years)	HR Model 1ª	HR Model 2 <sup>b</sup>	HR Model 3 <sup>c</sup>	<i>p</i> value Model 3 <sup>c</sup>
Cardiovascular outcome	0 1–4	20 17	4,442 1,027	4.5 16.6	Reference 3.61 (1.89–6.88)	Reference 2.01 (1.00–4.04)	Reference 2.12 (1.05–4.27)	- 0.049
	>4	21	894	23.5	5.05 (2.73–9.31)	2.38 (1.20–4.73)	2.38 (1.16–4.88)	0.028
		HR incr	ease per 1	AACS	1.13 (1.07–1.20)	1.07 (1.02–1.13)	1.08 (1.02–1.14)	0.002
All-cause mortality	0 1–4	12 9	4,553 1,121	2.6 8.0	Reference 3.06 (1.29–7.26)	Reference 2.08 (0.81–5.34)	Reference 2.13 (0.82–5.51)	_ 0.163
	>4	11	999	11.0	4.18 (1.84–9.47)	2.17 (0.85–5.52)	1.89 (0.70–5.06)	0.179
		HR incr	ease per 1	AACS	1.12 (1.07–1.19)	1.04 (0.95–1.11)	1.05 (0.95–1.18)	0.109
Renal outcome	0 1–4	96 30	4,443 1,105	21.6 27.1	Reference 1.23 (0.82–1.85)	Reference 1.20 (0.77–1.88)	Reference 1.29 (0.81–2.04)	_ 0.282
	>4	33	944	35.0	1.35 (0.91–2.01)	1.17 (0.74–1.83)	1.20 (0.75–1.92)	0.440
		HR incr	ease per 1	AACS	1.04 (0.95–1.11)	1.03 (0.92–1.13)	1.02 (0.96–1.08)	0.569

Table 4. Incidence and HRs for the outcomes according to AACS

The HR (95% CI) is reported for each model. The cardiovascular outcome was defined as myocardial infarction, coronary revascularization, stroke, and new-onset or aggravation of congestive heart failure. The renal outcome was defined as the first occurrence of a more than 50% decline in eGFR compared to a baseline value measured less than 28 days after kidney transplantation, or death-censored graft loss requiring maintenance dialysis for more than 3 months, or re-transplantation. AACS, abdominal aortic calcification score; HR, hazard ratio. <sup>a</sup>Model 1: a crude analysis without adjustment. <sup>b</sup>Model 2: adjusted for CCI, smoking status, education levels, body mass index, and cause of end-stage kidney disease. <sup>c</sup>Model 3: model 2 + type of donor (living or deceased), duration of dialysis before kidney transplantation, desensitization, and immunosuppression medications.

Similarly, according to the log-rank test, no significant difference was observed in renal outcomes among the different AACS groups (p = 0.286, Fig. 3c). HR for the high AACS group compared to the low AACS group was 1.20 (95% CI: 0.75–1.92, p = 0.440, model 3, Table 4). When CACS and AACS were treated as continuous variables, the HR increase per 1 CACS was calculated as 1,000 (95% CI: 0.994–1.004, p = 0.166, model 3) in the multivariate model (Table 3). The HR increase per 1 AACS was 1.02 (95% CI: 0.96–1.08, p = 0.569, model 3) in the multivariate model (Table 4).

# Prediction Models Based on CACS or AACS

When CACS or AACS was added to the respective base models for major outcomes, significant NRI scores were observed (online suppl. Table S4). NRI was 0.636 (95% CI: 0.067–0.967, p = 0.006) when CCAS was added to the

base model to predict cardiovascular outcomes. Moreover, NRI was 0.336 (95% CI: -0.060 - 0.719, p = 0.091) when AACS was added to the base model to predict cardiovascular outcomes. The Cox regression model for the composite outcome of cardiovascular outcome and all-cause mortality showed an NRI of 0.700 (95% CI: 0.085–1.018, p = 0.003) compared to the base model. No statistically significant improvements were observed in C-statistics.

# Discussion

This multicenter KT cohort study investigated the association between two well-known vascular calcification markers, CACS/AACS, and posttransplant outcomes. This



Downloaded from http://karger.com/kdd/article-pdf/10/4/249/4263801/000538929.pdf by Keimyung University user on 11 December 2024

Fig. 3. Kaplan-Meier curves for incidence of cardiovascular outcomes (a), all-cause mortality (b), and renal outcomes (c) according to AACS groups (AACS = 0; AACS = 1–4, AACS > 4). *p* < 0.001, 0.001, and 0.286 for cardiovascular outcomes, all-cause mortality, and renal outcomes, respectively (log-rank test). AACS, aortic artery calcification score.

### Ha/Jeong/Ryu/Kim/Huh/Lee/Jung/Kang/ Ro/Han/Seok Kim/Yang

demonstrates that higher levels of CACS and AACS were associated with an increased risk for cardiovascular events and mortality in KT patients; however, neither was associated with the risk for renal function deterioration in KT patients.

Both CACS and AACS have increased over posttransplant 5 years and progression of vascular calcification seemed to be more prominent in the low and intermediate CACS/AACS subgroups. A previous study showed that old age and high FGF23 levels were common risk factors for CACS and AACS progression in dialysis patients [31], while we found that old age and DM were common risk factors for the progression of CACS and AACS in KT patients.

A Korean multicenter cohort study known as KNOW-CKD showed that high CACS was independently associated with adverse cardiovascular outcomes and allcause mortality in CKD stages G1-G5 without kidney replacement therapy [15]. A single-center study also showed that high AACS based on CT was associated with an increased risk for CVD in patients with CKD [17]. In the KT population, several small-sized, singlecenter studies reported that higher CACS or AACS was associated with an increased risk for CVD, similar to the CKD population [20–24]. This multicenter study demonstrated that higher CACS and AACS were associated with an increased risk for CVD, and it strengthened the association of vascular calcification with CVD in KT patients using CACS and AACS simultaneously in a single study. These findings suggest that pretransplant CACS and AACS could be potential prognostic markers for posttransplant CVD in KT patients as well as patients with CKD. Further, we may be able to identify high-risk KT patients that could benefit from receiving better monitoring and intervention using these scores.

The impact of CACS on renal disease progression in patients with CKD has been controversial. Some smallsized studies reported a significant association between CACS and renal function deterioration, whereas other small-sized studies did not show this significant association in patients with CKD [32–34]. Recently, KNOW-CKD, a multicenter study with larger sample size, showed that high CACS was significantly associated with deterioration in kidney function for all stages of CKD without renal replacement therapy [16]. Although there were limited studies on the association between AACS and renal functions, a cross-sectional study reported a significant association between AACS and eGFR in patients with CKD [18]. Several possible mechanisms have been proposed to explain the association between vascular calcification and CKD progression, such as direct renal vascular calcification or renal function decline secondary to cardiorenal syndrome after cardiovascular compromise; however, the dominant mechanism is yet unknown [32, 33].

To the best of our knowledge, our study is the first study to analyze the association of CACS and AACS with renal function progression in KT patients. Contrary to CVD, we found no significant association between vascular calcification scores and renal outcomes. The lack of association with renal outcomes was unexpected, as vascular calcification and CKD progression are known to share many risk factors. One possible explanation for this negative finding is that KT patients have distinct pathophysiological mechanisms from those with CKD, which cause renal function decline and graft loss. Various KT-specific factors, including organ donor quality, human leukocyte antigen matching, sensitization status, and immunosuppressive regimens, may influence graft functional deterioration in KT patients and mitigate the impact of vascular calcification on kidney functional deterioration [35]. However, the statistical power of our study may be insufficient to detect a significant association between arterial calcification and renal outcomes in KT patients. Further studies with larger sample sizes and longer follow-up periods are required to clarify this issue.

CACS improved the predictive performance for cardiovascular outcomes in the CKD population [15, 36]. Because we found that CACS and AACS were significantly associated with cardiovascular outcomes, we investigated whether the addition of CACS or AACS to conventional prediction models can improve the predictive performance for cardiovascular outcomes. This study demonstrated that the addition of CACS led to the modest improvement in the predictive performance for cardiovascular outcomes, and the addition of AACS also showed a trend of improved prediction performance, supporting the clinical usefulness of CACS and AACS in the KT population.

This study has several strengths that contribute substantially to our understanding of the role of CACS and AACS in post-KT CVD and renal outcomes. First, this is the first multicenter cohort study with the largest sample size and relatively long-term follow-up duration in the KT population, in contrast to previous small-sized, single-center studies. Second, although previous studies have analyzed the impact of either CACS or AACS alone on CVD in the KT population, we simultaneously analyzed the impact of both CACS and AACS on renal outcomes and CVD, providing a more holistic view. Third, we proposed prediction models and performed a risk analysis for CVD and renal outcomes, which was in contrast to previous studies on KT patients. However, this study has several limitations. First, we did not perform a validation study in the external validation group. Second, despite the large sample size and long-term follow-up duration, the number of events was still relatively small, which might have led to failure in detecting significant association between CACS/AACS and renal outcomes. Future studies for other populations could confirm and generalize our findings.

In conclusion, pretransplant arterial calcification, characterized by high CACS or AACS, is an independent risk factor for posttransplant CVD and mortality but not for graft dysfunction in KT patients. Our findings suggest a potential role of pretransplant screening of CACS and AACS in risk stratification for post-KT outcomes.

# Acknowledgments

We thank all study investigators and coordinators who participated in the KNOW-KT.

# Statement of Ethics

This study protocol was reviewed and approved by the Regional Ethics Committee and Institutional Review Board of Severance Hospital, Yonsei University College of Medicine

### References

- 1 Rangaswami J, Mathew RO, Parasuraman R, Tantisattamo E, Lubetzky M, Rao S, et al. Cardiovascular disease in the kidney transplant recipient: epidemiology, diagnosis and management strategies. Nephrol Dial Transplant. 2019;34(5):760–73. https://doi. org/10.1093/ndt/gfz053.
- 2 Kasiske BL, Guijarro C, Massy ZA, Wiederkehr MR, Ma JZ. Cardiovascular disease after renal transplantation. J Am Soc Nephrol. 1996;7(1):158–65. https://doi.org/10. 1681/ASN.V71158.
- 3 Birdwell KA, Park M. Post-transplant cardiovascular disease. Clin J Am Soc Nephrol. 2021;16(12):1878–89. https://doi.org/10.2215/ CJN.00520121.
- 4 Korogiannou M, Xagas E, Marinaki S, Sarafidis P, Boletis JN. Arterial stiffness in patients with renal transplantation; associations with Co-morbid conditions, evolution, and prognostic importance for cardiovascular and renal outcomes. Front

Cardiovasc Med. 2019;6:67. https://doi.org/ 10.3389/fcvm.2019.00067.

- 5 Kim JS, Hwang HS. Vascular calcification in chronic kidney disease: distinct features of pathogenesis and clinical implication. Korean Circ J. 2021;51(12):961–82. https://doi.org/ 10.4070/kcj.2021.0995.
- 6 Henaut L, Chillon JM, Kamel S, Massy ZA. Updates on the mechanisms and the care of cardiovascular calcification in chronic kidney disease. Semin Nephrol. 2018;38(3):233–50. https://doi.org/10.1016/j.semnephrol.2018. 02.004.
- 7 Yang H, Curinga G, Giachelli CM. Elevated extracellular calcium levels induce smooth muscle cell matrix mineralization in vitro. Kidney Int. 2004;66(6):2293–9. https://doi. org/10.1111/j.1523-1755.2004.66015.x.
- 8 Jono S, Shioi A, Ikari Y, Nishizawa Y. Vascular calcification in chronic kidney disease. J Bone Miner Metab. 2006;24(2):176–81. https://doi.org/10.1007/s00774-005-0668-6.

(4-2012-0223, 4-2023-0758). This full list of participating sites and ethics committees can be found at Supplementary Information. All patients enrolled in KNOW-KT voluntarily provided written informed consent.

# **Conflict of Interest Statement**

The authors of this manuscript have no conflicts of interest to disclose.

# Funding Sources

This work was supported by Research Programs funded by the Korea Disease Control and Prevention Agency (2011E3300300, 2012E3301100, 2013E3301600, 2013E3301601, 2013E3301602, 2016E3300200, 2016E3300201, 2019E320100, 2019E320101, 2019E320102, 2022-11-007).

# **Author Contributions**

Research design and writing: J.H. and J.Y. Data collection: J.H., J.-H.R., M.-G.K., K.H.H., K.W.L., H.-Y.J., K.P.K., H.R., S.H., B.S.K., and J.Y. Data analysis and data interpretation: J.H., J.C.J., and J.Y.

# Data Availability Statement

Data supporting the findings of this study are not publicly available due to privacy reasons but are available from the data sharing committee upon reasonable request.

- 9 Shigematsu T, Kono T, Satoh K, Yokoyama K, Yoshida T, Hosoya T, et al. Phosphate overload accelerates vascular calcium deposition in endstage renal disease patients. Nephrol Dial Transplant. 2003;18(Suppl 3):iii86–89. https:// doi.org/10.1093/ndt/gfg1022.
- 10 An C, Lee HJ, Lee HS, Ahn SS, Choi BW, Kim MJ, et al. CT-based abdominal aortic calcification score as a surrogate marker for predicting the presence of asymptomatic coronary artery disease. Eur Radiol. 2014;24(10):2491–8. https:// doi.org/10.1007/s00330-014-3298-3.
- 11 Agatston AS, Janowitz WR, Hildner FJ, Zusmer NR, Viamonte M Jr, Detrano R. Quantification of coronary artery calcium using ultrafast computed tomography. J Am Coll Cardiol. 1990;15(4):827–32. https://doi. org/10.1016/0735-1097(90)90282-t.
- 12 Demer LL, Tintut Y. Vascular calcification: pathobiology of a multifaceted disease. Circulation. 2008;117(22):2938–48. https://doi.org/ 10.1161/CIRCULATIONAHA.107.743161.

- 13 Kauppila LI, Polak JF, Cupples LA, Hannan MT, Kiel DP, Wilson PW. New indices to classify location, severity and progression of calcific lesions in the abdominal aorta: a 25-year follow-up study. Atherosclerosis. 1997; 132(2):245–50. https://doi.org/10.1016/s0021-9150(97)00106-8.
- 14 Vannini FC, Teixeira AS, Caramori JC, Martin LC, Barretti P. Is Kauppila method able to detect the progression of vascular calcification and predict cardiovascular events in patients undergoing hemodialysis? Clin Nephrol. 2016;85(2):84–91. https://doi. org/10.5414/cn108578.
- 15 Jung CY, Yun HR, Park JT, Joo YS, Kim HW, Yoo TH, et al. Association of coronary artery calcium with adverse cardiovascular outcomes and death in patients with chronic kidney disease: results from the KNOW-CKD. Nephrol Dial Transplant. 2023;38(3):712–21. https://doi. org/10.1093/ndt/gfac194.
- 16 Yun HR, Joo YS, Kim HW, Park JT, Chang TI, Son NH, et al. Coronary artery calcification score and the progression of chronic kidney disease. J Am Soc Nephrol. 2022;33(8):1590–601. https:// doi.org/10.1681/ASN.2022010080.
- 17 Tatami Y, Yasuda Y, Suzuki S, Ishii H, Sawai A, Shibata Y, et al. Impact of abdominal aortic calcification on long-term cardio-vascular outcomes in patients with chronic kidney disease. Atherosclerosis. 2015; 243(2):349–55. https://doi.org/10.1016/j. atherosclerosis.2015.10.016.
- 18 Zhou Y, Hellberg M, Kouidi E, Deligiannis A, Hoglund P, Clyne N. Relationships between abdominal aortic calcification, glomerular filtration rate, and cardiovascular risk factors in patients with non-dialysis dependent chronic kidney disease. Clin Nephrol. 2018;90(6):380–9. https://doi.org/ 10.5414/CN109441.
- 19 Podesta MA, Cucchiari D, Ciceri P, Messa P, Torregrosa JV, Cozzolino M. Cardiovascular calcifications in kidney transplant recipients. Nephrol Dial Transplant. 2022;37(11): 2063–71. https://doi.org/10.1093/ndt/gfab053.
- 20 Roe P, Wolfe M, Joffe M, Rosas SE. Inflammation, coronary artery calcification and cardiovascular events in incident renal transplant recipients. Atherosclerosis. 2010;

212(2):589–94. https://doi.org/10.1016/j. atherosclerosis.2010.05.016.

- 21 Nguyen PT, Henrard S, Coche E, Goffin E, Devuyst O, Jadoul M. Coronary artery calcification: a strong predictor of cardiovascular events in renal transplant recipients. Nephrol Dial Transplant. 2010;25(11): 3773–8. https://doi.org/10.1093/ndt/gfq268.
- 22 DeLoach SS, Joffe MM, Mai X, Goral S, Rosas SE. Aortic calcification predicts cardiovascular events and all-cause mortality in renal transplantation. Nephrol Dial Transplant. 2009;24(4): 1314–9. https://doi.org/10.1093/ndt/gfn753.
- 23 Claes KJ, Viaene L, Heye S, Meijers B, d'Haese P, Evenepoel P. Sclerostin: another vascular calcification inhibitor? J Clin Endocrinol Metab. 2013;98(8):3221–8. https:// doi.org/10.1210/jc.2013-1521.
- 24 Seyahi N, Alagoz S, Atli Z, Ozcan SG, Tripepi G, Bakir A, et al. Coronary artery calcification progression and long-term cardiovascular outcomes in renal transplant recipients: an analysis by the joint model. Clin Kidney J. 2022;15(1):101–8. https://doi.org/10.1093/ ckj/sfab174.
- 25 Yang J, Lee J, Huh KH, Park JB, Cho JH, Lee S, et al. KNOW-KT (Korean cohort study for outcome in patients with kidney transplantation: a 9-year longitudinal cohort study): study rationale and methodology. BMC Nephrol. 2014;15:77. https://doi.org/10.1186/ 1471-2369-15-77.
- 26 Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis. 1987; 40(5):373–83. https://doi.org/10.1016/0021-9681(87)90171-8.
- 27 Ma D, Yan H, Yang X, Yu Z, Ni Z, Fang W. Abdominal aortic calcification score as a predictor of clinical outcome in peritoneal dialysis patients: a prospective cohort study. BMC Nephrol. 2020;21(1):151. https://doi. org/10.1186/s12882-020-01822-9.
- 28 Levey AS, Stevens LA, Schmid CH, Zhang YL, Castro AF 3rd, Feldman HI, et al. A new equation to estimate glomerular filtration rate. Ann Intern Med. 2009;150(9):604–12. https://doi.org/10.7326/0003-4819-150-9-200905050-00006.

- 29 White IR, Royston P, Wood AM. Multiple imputation using chained equations: issues and guidance for practice. Stat Med. 2011; 30(4):377–99. https://doi.org/10.1002/sim. 4067.
- 30 Cook NR. Quantifying the added value of new biomarkers: how and how not. Diagn Progn Res. 2018;2:14. https://doi.org/10. 1186/s41512-018-0037-2.
- 31 Zhang H, Li G, Yu X, Yang J, Jiang A, Cheng H, et al. Progression of vascular calcification and clinical outcomes in patients receiving maintenance dialysis. JAMA Netw Open. 2023;6(5):e2310909. https://doi.org/10.1001/jamanetworkopen. 2023.10909.
- 32 Lamarche MC, Hopman WM, Garland JS, White CA, Holden RM. Relationship of coronary artery calcification with renal function decline and mortality in predialysis chronic kidney disease patients. Nephrol Dial Transplant. 2019;34(10):1715–22. https://doi. org/10.1093/ndt/gfy183.
- 33 Garland JS, Holden RM, Hopman WM, Gill SS, Nolan RL, Morton AR. Body mass index, coronary artery calcification, and kidney function decline in stage 3 to 5 chronic kidney disease patients. J Ren Nutr. 2013; 23(1):4–11. https://doi.org/10.1053/j.jrn. 2011.12.008.
- 34 Dellegrottaglie S, Saran R, Gillespie B, Zhang X, Chung S, Finkelstein F, et al. Prevalence and predictors of cardiovascular calcium in chronic kidney disease (from the Prospective Longitudinal RRI-CKD Study). Am J Cardiol. 2006;98(5):571–6. https://doi.org/10.1016/j. amjcard.2006.03.030.
- 35 Pratschke J, Weiss S, Neuhaus P, Pascher A. Review of nonimmunological causes for deteriorated graft function and graft loss after transplantation. Transpl Int. 2008;21(6): 512–22. https://doi.org/10.1111/j.1432-2277. 2008.00643.x.
- 36 Chen J, Budoff MJ, Reilly MP, Yang W, Rosas SE, Rahman M, et al. Coronary artery calcification and risk of cardiovascular disease and death among patients with chronic kidney disease. JAMA Cardiol. 2017;2(6): 635–43. https://doi.org/10.1001/jamacardio. 2017.0363.